Author’s response to reviews


Authors:

Flavio Cadegiani (f.cadegiani@gmail.com)
Claudio Kater (claudio.kater@gmail.com)

Version: 3 Date: 11 Oct 2019

Author’s response to reviews:

Editor Comments:

1. Please change the corresponding email address in the manuscript so that it matches the corresponding email address provided on the editorial manager system.

R. Changed accordingly.

2. Please format your abstract under the following headings:
   • Background
   • Methods
   • Results
   • Conclusions
   • Keywords

R. Changed accordingly.

3. Please amend the statement in the "Consent for publication" section, as this refers to consent obtained to publish potentially-identifying information (of patients/participants). If your manuscript does not contain any individual person’s identifiable data or information, please state “Not applicable” under this section.
R. Changed accordingly.

4. Please separate the "Availability of data and materials and funding" into two sections, titled "Availability of data and materials" and "Funding".

R. Separated accordingly.

5. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

R. We double checked, and now all abbreviations are defined in the first use in the text.
List of abbreviations is now provided.

6. The tables are not cited in the body of the manuscript, please amend this.

R. Table 6 was not cited in the body of the manuscript.
It is now described in “line 320” (“Table 6 depicts the differences between the level of hormonal responsiveness to ITT between ATL and NPAC.”)

7. Figure 4 does not have a legend, please amend this.

R. Amended accordingly.

8. Please remove the response to reviewers additional file, as it is no longer required at this stage.

R. Ok.

9. Please ensure that the manuscript only has one title.
R. The title is: “Enhancement of hypothalamic-pituitary activity in male athletes: evidence of a novel hormonal mechanism of physical conditioning.” And has been corrected where needed.

10. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.